Author's response to reviews

Title: high prevalence of chronic kidney disease in Iran: a large population-based study

Authors:

Farhad Hosseinpanah (fhospanah@endocrine.ac.ir)
Farshad Kasraei (farshad_ka@yahoo.com)
Amir A Nassiri (nassiri@ams.ac.ir)
Fereidoun Azizi (azizi@endocrine.ac.ir)

Version: 4 Date: 19 January 2009

Author's response to reviews:

Dear editors of BMC Public Health

Thank you very much for accepting our article to be published in your journal. we have made the changes you asked to the format of our manuscript, and herewith we submit the revised version.

Farhad Hosseinpanah